Author’s response to reviews

Title: Relationship between inflammatory markers and mild cognitive impairment in Chinese patients with type 2 diabetes: a case-control study

Authors:
Miaoyan Zheng (myzheng73@163.com)
Baocheng Chang (changbc1970@126.com)
Liqiang Tian (defjmmm@163.com)
Chunyan Shan (chunyanshan@hotmail.com)
Hui Chen (602090529@qq.com)
Yuxia Gao (gaoyuxia@medmail.com.cn)
Guowei Huang (tjgwhuang@126.com)
Meilin Zhang (zhangmeilin@tmu.edu.cn)

Version: 6 Date: 24 Jun 2019

Author’s response to reviews:

Dear Professor Ciarán Martin Fitzpatrick,

Thank you for the opportunity to improve our manuscript (BEND-D-19-00016) \(^1\). We have revised the manuscript carefully according to the comments of the reviewers. We provide below our detailed responses to your comments.

If you have any questions about our revised manuscript, please inform us without hesitation.

Sincerely Yours,

Meilin Zhang
Department of Nutrition and Food Science
School of Public Health
Tianjin Medical University
The point-by-point responses to the reviewers’ comments below:

Editor Comments:

1. Figure 1
We note you have cited figure 1 in your manuscript but no such figure has been included. Please ensure any figures are uploaded as separate files along with your manuscript.
Response: Thank you for your consideration. We have uploaded figure 1 as separate file.

2. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Please remove any files that should not be published along with your manuscript e.g. cover letters, reviewer responses, guidelines etc., ready for publication.
Response: We have uploaded the revised manuscript as a single, final, clean version.